Author's response to reviews

Title: Corneal Collagen Cross-linking in a late-onset graft infectious ulcer: a case report.

Authors:

Georgios Labiris (labiris@usa.net)
Athanassios Giarmoukakis (thangiar@googlemail.com)
Roman Larin (larinro@yahoo.co.uk)
Haris Sideroudi (harissid@alex.duth.gr)
Vassilios Kozobolis (vkozompo@med.duth.gr)

Version: 3
Date: 28 January 2014

Author's response to reviews: see over
Dear editor,

We appreciate your effort in reviewing our manuscript. Please find below our response to your comments.

On behalf of all authors

Dr Georgios Labiris

1. Include authors’ emails. Emails included accordingly.
2. Competing interests repositioning. Repositioned accordingly.
4. Remove hospital name. Done.
5. Include authors contribution. Done.
6. Acknowledgements. None.
7. Figure title repositioning. Done.

Reviewer 1.
We appreciate your effort in reviewing our manuscript.

Reviewer 2.
We appreciate your effort in reviewing our manuscript.

1. Substitute “of” with “for”. Done.
2. Substitute “12” with “twelve”. Done.
3. Demarcation line. Dear reviewer, the exact localisation of the demarcation line is almost impossible due to the severe interference in the OCT images by the underlying pathology at the cornea. Therefore, we apologise but we cannot comply with this comment since we are not 100% sure on the exact localisation.

Reviewer 3.
We appreciate your effort in reviewing our manuscript.

1. Spelling review. Spelling was done again.
2. Cite references. References cited as suggested.
3. Highlight MRSA as a prevalent post-PK infection. Please refer to Discussion / Page 6 / Line 7 / “Staphylococcus aureus is among the most common … to prevent rejection and infection.”